Author's response to reviews

Title: Effect of exercise therapy on lipid profile and oxidative stress indicators in patients with type 2 diabetes

Authors:

Lorenzo Gordon (lorenzogordon@yahoo.com)
Errol Y Morrison (errol.morrison@uwimona.edu.jm)
Donovan A McGrowder (dmcgrowd@yahoo.com)
Ronald Young (ronald.young@uwimona.edu.jm)
Yeiny P Fraser (yeinita@yahoo.com)
Eslaen M Zamoraz (emortorellz@yahoo.es)
Ruby L Alexander-Lindo (lisa.lindo@uwimona.edu.jm)
Rachael Irving (rachael.irving@uwimona.edu.jm)

Version: 4 Date: 22 April 2008

Author's response to reviews: see over
April 22, 2008

Iratxe Puebla,
Senior Assistant Editor
BMC-series journals

Tel: +44 (0)20 7631 9921
Facsimile: +44 (0)20 7631 9923
e-mail: editorial@biomedcentral.com
Web: http://www.biomedcentral.com

Dear Madam,

On behalf of the authors of the manuscript I wish to thank you for your assistance, patience and for so willing accommodating us. In response to your e-mail on February 19, 2008, we have made all the changes as recommended. The trial was registered in the Australian New Zealand Clinical Trials Registry (ANZCTR) and allocated the ACTRN: ACTRN12608000217303. The address for the trial is http://www.ANZCTR.org.au/ACTRN12608000217303.aspx

The Background section was elaborated and the word count for the abstract is 349. The acknowledgement included cited persons other than the authors who have contributed to the study, and those who gave financial assistance. The figures will be upgraded separately from the manuscript as recommended.

Thanks again for kind consideration and assistance.

Yours sincerely,

Donovan McGrowder, MSc, PhD, FIBMS
Lecturer in Chemical Pathology
Department of Pathology
University of The West Indies